Author's response to reviews

Title: Assessing the accessibility of HIV care packages among tuberculosis patients in the Northwest Region, Cameroon

Authors:

Nwarbébé Barnabas Njozing (njozing.barnabas@epiph.umu.se)
Miguel San Sebastian (miguel.sansebastian@epiph.umu.se)
Pius Muffih Tih (piustih@cbchealthservices.org)
Anna-Karin Hurtig (anna-karin.hurtig@epiph.umu.se)

Version: 8 Date: 10 March 2010

Author's response to reviews:

1. In title, "North-West Region, Cameroon" has been changed to "Northwest Region, Cameroon"

2. On the penultimate sentence on page 9, paragraph 2, "Diagnosis and treatment is..." has been changed to "Diagnosis and treatment are..."

3. Minor adjustments on table 1 (adjusting the positioning of the frequency of patients for the hospital BRH) and table 2 (including the closing bracket in the confidence interval for the multivariate analysis of the age group 15-29 years)